Author's response to reviews

Title: Traumatic pulmonary pseudocysts: two case reports

Authors:

Bulent Kocer (drbkocer@gmail.com)
Gultekin Gulbahar (mdgultekin@gmail.com)
Nesimi Gunal (nesimigunal@yahoo.com.tr)
Koray Dural (koraydural@yahoo.com)
Unal Sakinci (unalsakinci@hotmail.com)

Version: 5 Date: 14 September 2007

Author's response to reviews:

1. The authors qualifications/titles removed from the manuscript.
2. Corresponding authors telephone and fax numbers are removed from the text.
3. Introduction -> Background
4. References cited in the text in square brackets.
5. We put Consent under the Acknowledgments header.
6. A Tables section title placed after the Figure legends